Author’s response to reviews

Title: Differentials in prevalence and correlates of metabolic risk factors of non-communicable diseases among women in Sub-Saharan Africa: Evidence from 33 countries

Authors:
Sanni Yaya (sanni.yaya@uottawa.ca)
Michael Ekholuenetale (mic42006@gmail.com)
Ghose Bishwajit (brammaputram@yahoo.fr)

Version: 3 Date: 26 Sep 2018

Author’s response to reviews:

1. Email addresses of all authors were included on the title page and we confirm they are identical to the ones entered on the submission system.

2. We checked authors' contributions and we can confirm they are in line with the ICMJE guidelines and that all listed authors have performed all four points specified in your note.

3. We indicated that Sanni Yaya is an Editorial Board member for BMC Public Health in the competing interest section.

4. We have uploaded our manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours.

Thanks